Author's response to reviews

Title: Terminal chromosome 4q deletion syndrome in an infant with hearing impairment and moderate syndromic features: review of literature

Authors:

Barbara Vona (barbara.vona@uni-wuerzburg.de)
Indrajit Nanda (nanda@biozentrum.uni-wuerzburg.de)
Cordula Neuner (cordula.neuner@uni-wuerzburg.de)
Joerg Schröder (joerg.schroeder@biozentrum.uni-wuerzburg.de)
Vera M Kalscheuer (kalscheu@molgen.mpg.de)
Waafa Shehata-Dieler (dieler_w@ukw.de)
Thomas Haaf (thomas.haaf@uni-wuerzburg.de)

Version: 3
Date: 19 June 2014

Author's response to reviews:

Dear Professor Neri,

We have incorporated all the changes, suggested in your Email of June 19 and hope that our manuscript is ready for publication now.

Thank you very much again for all the efforts with of our manuscript.

Sincerely yours,

Thomas Haaf

1) on p. 4, the expression "with an uncomplicated pregnancy" was changed to "after an uncomplicated pregnancy"

2) on p. 5, the expression "he was euthyroid" was changed to "he had normal thyroid function"

3) on p. 12, the expression "the majority of this interval" was changed to "the major part of this interval"